Author’s response to reviews

Title: HOPE: Help fOr People with money, employment, benefit or housing problems: study protocol for a randomised controlled trial.

Authors:

maria barnes (maria.barnes@bristol.ac.uk)
Anne Haase (anne.hasse@bristol.ac.uk)
Alison Bard (alison.bard@bristol.ac.uk)
Jenny Donovan (Jenny.Donovan@bristol.ac.uk)
Rosemary Davies (Rosemary3.Davies@uwe.ac.uk)
Sue Dursley (Sue.Dursley@nhs.net)
John Potokar (John.Potokar@bristol.ac.uk)
Navneet Kapur (nav.kapur@manchester.ac.uk)
Keith Hawton (keith.hawton@psych.ox.ac.uk)
Rory O'Connor (rory.oconnor@glasgow.ac.uk)
William Hollingworth (william.hollingworth@bristol.ac.uk)
Chris Metcalfe (chris.metcalfe@bristol.ac.uk)
David Gunnell (D.J.Gunnell@bristol.ac.uk)

Version: 2 Date: 07 Aug 2017

Author’s response to reviews:

With apologies, I have now attached the updated flow chart as requested.

Best wishes

Maria